Author's response to reviews

Title: Increased PADI4 expression in blood and tissues of patients with malignant tumors

Authors:

Xiaotian Chang (changxt@126.com)
Jinxiang Han (jinxiang.han@163.com)
Li Pang (lipang2h@163.com)
Yan Zhao (zhaoyansdams@163.com)
Yi Yang (yiyang@163.com)
Zhonglin Shen (zhonglinshen@163.com)

Version: 4 Date: 23 January 2009

Author's response to reviews:

Dear editor:
Thank you very much for you to accept our submission. According to your instructions, we made following point-to-point corrections with our paper.

1. about author list.
We carefully checked the list of authors which is written identical to that in the submission system.

2. about figure legends
Figure 1 contained 24 pictures which were put on 2 pages (figure 1-1 and figure 1-2) and were submitted separately.

4. We added email addresses for all the authors on the title page by the format editor required.

5. We renamed Materials & Methods to 'Methods' in the resubmission.

6. We carefully checked manuscript and corrected typographical errors in the resubmission.
Thank you very much!
With best regards!
Yours,
Xioatian Chang